#### **RESEARCH ARTICLE**

# Explaining the allocation of legislative specialization

## Aldo F. Ponce 厄

Department of Political Studies, Centro de Investigación y Docencia Económicas, Mexico City, Mexico Email: aldo.ponce@cide.edu

(Received 1 November 2022; revised 16 May 2023; accepted 20 July 2023; first published online 18 October 2023)

## Abstract

This article explains why there is important variation in the degree of concentration of legislative specialization across legislative parties. Greater concentration of legislative specialization leads parties to concentrate their legislative efforts on a smaller set of policy jurisdictions. Through enhancing their concentration of legislative specialization in certain policy areas, parties might more clearly signal their policy concerns and interests to voters. This study argues and shows that party size alters the concentration of legislative specialization. In particular, I find that a U-shaped curve defines the relationship between party size and the degree of concentration of legislative specialization: niche parties and the largest parties choose higher levels of concentration of legislative specialization.

Keywords: concentration of legislative specialization; Latin America; party size

## 1. Introduction

This article explains why there is important variation in the degree of concentration of legislative specialization across legislative parties. I define legislative specialization at the party level as the proportion of legislative effort parties make by each policy jurisdiction. For instance, if a legislative party spent more than half of its available resources (e.g. time, staff) on education and health, we could state that this party specializes relatively more on these two policy jurisdictions than on other areas. Greater concentration of legislative specialization implies that parties allocate their legislative efforts on a smaller set of policy jurisdictions. As I discuss below, party leaders often promote and coordinate such efforts.<sup>1</sup> As party leaders are able to coordinate legislative efforts, they might prefer increasing the concentration of legislative specialization to signal preferences, interest, or effort, and eventually either achieve electoral benefits due to issue voting in the short term or construct the party label in the long term.

Previous studies argue that delegation of legislative functions to specialized committees in charge of specific policy jurisdictions occurs for three reasons. First, under the informational perspective, legislative majorities delegate prerogatives to committees because they need expertise and greater information on certain policy jurisdictions (Gilligan and Krehbiel, 1987; Krehbiel, 1991; Hamm *et al.*, 2011). As members of these specialized committees share their policy expertise, legislatures accomplish collective benefits from specialization (Krehbiel, 1991). Second, under the distributional perspective, special privileges are granted to committees to solve the problem of majority rule instability and to facilitate log rolls (Weingast and Marshall, 1988; Baron and Ferejohn, 1989). Third, partisan theories

<sup>&</sup>lt;sup>1</sup>Legislators might also specialize in certain policy jurisdictions (the committees in which they work) independently of the agenda party leaders pursue within congress. Hence, the concept of legislative specialization at the party level is different from the type of specialization that individual legislators develop when concentrating their efforts on policy jurisdictions for which they often possess policy expertise. I also highlight that specialization in this study does not mean policy expertise.

<sup>©</sup> The Author(s), 2023. Published by Cambridge University Press

contend that delegation to committees occurs to strengthen the control of party leaders over the legislative agenda (Rohde, 1991; Cox and McCubbins, 1993).

Although these approaches offer convincing explanations as to why legislative specialization occurs through the delegation of rights and prerogatives to committees, there is still scant research on how legislative parties allocate their policy initiatives across committees. The existence of differing electoral rewards across policy jurisdictions can justify why the distribution of efforts is uneven. In fact, party attention varies across policy jurisdictions, prioritizing salient policy jurisdictions that provide greater electoral rewards to parties. However, this study does not aim to inform which policy jurisdictions are preferred by parties. Instead, it explains why there is variation in the concentration of legislative specialization across parties. To answer this question, this article focuses on the key role of party size measured by the percentage of seats held in congress. This factor is hypothesized to alter the benefits and costs of the concentration of legislative specialization. By employing a unique database composed by 66,027 bill initiatives from 10 Latin American legislatures, I find that a U-shaped curve defines the relationship between party size and the degree of concentration of legislative specialization: niche parties and the largest parties tend to choose the highest degrees of concentration of legislative specialization.

Greater concentration of legislative specialization might bring parties either short-term or longterm benefits. In the short term, showing policy preferences in highly salient policy jurisdictions might help parties gain support, especially electoral advantages during campaigns (Rabinowitz *et al.*, 1982; Budge and Farlie, 1983; Edwards *et al.*, 1995; Fournier *et al.*, 2003; Bélanger and Meguid, 2008; Walgrave *et al.*, 2012; Klüver and Sagarzazu, 2016; Schröder and Stecker, 2018). Some studies report that issue voting is gaining increasing importance when explaining citizens' voting decisions (Lefevere *et al.*, 2015; Bos *et al.*, 2016).<sup>2</sup> Further research is needed to determine to what extent greater concentration of legislative specialization helps parties achieve greater electoral support by targeting highly salient issues.

In the long term, as political parties consistently specialize in certain policy jurisdictions in the legislature, they constantly inform society about their policy preferences (Green-Pedersen, 2010; Green-Pedersen and Mortensen, 2015; Schröder and Stecker, 2018). Over time, greater concentration of legislative specialization might help them construct a reputation as a competent party to deal with problems regarding a policy jurisdiction, and perhaps, such a reputation could facilitate an ownership claim on these policy issues. Walgrave *et al.* (2012) state that issue ownership refers to 'the fact that specific political parties are identified by the public at large with specific policy issues and are considered to be the most competent party to deal with those issues'. Parties take advantage of issue ownership to enhance the support of the electorate in future elections, since ownership indicates which problems the party will address (Budge and Farlie, 1983; Petrocik, 1996; Nadeau *et al.*, 2001; Aldrich and Griffin, 2003; Petrocik *et al.*, 2003; Clarke *et al.*, 2004; Damore, 2004; Holian, 2004; van der Brug, 2004; Hayes, 2005; Green and Hobolt, 2008; Neundorf and Adams, 2018; Green and Jennings, 2019). Further research is needed to reveal to what extent greater concentration of legislative specialization helps parties acquire issue ownership or a reputation as the most competent party in specific policy jurisdictions.

Finally, greater concentration of legislative specialization in certain policy domains can provide voters with greater information on parties' preferences and goals; and it helps them distinguish parties based on policy domains. If voters can distinguish across political parties according to the policy domains voters believe parties prioritize, then these parties could enhance their 'programmaticness'. Since programmaticness is one of the characteristics of 'institutionalized' parties and party systems (Mainwaring, 2018), greater concentration of legislative specialization might also contribute to the formation of institutionalized parties and party systems. Further research is also needed to verify this potential consequence.

<sup>&</sup>lt;sup>2</sup>To enhance support, parties deliberately target the media to inform the public about their actions in legislatures (Norris *et al.*, 1999; Walgrave and De Swert, 2004, 2007; Walgrave *et al.*, 2009; Schröder and Stecker, 2018).

The rest of this article is divided into four sections. Section 2 explains the hypothesis of this study. Section 3 discusses how other studies have characterized the role of party leaders in coordinating legislative actions in Latin American legislatures. Section 4 presents empirical evidence regarding the impact of party size on the degree of concentration of legislative specialization. Section 5 concludes and offers suggestions for further research.

#### 2. Explaining the degree of concentration of legislative specialization

This section explains why parties choose greater or lesser degrees of concentration of legislative specialization. In this analysis, I treat the legislative party as the unit of analysis that makes decisions to maximize electoral support. I assume that legislative parties could win seats by either responding to the demands of local constituencies or of the population at large.

Moreover, for simplicity I assume that the number of bill initiatives serves as a good proxy of legislative effort. Bill initiation data can indicate the types of policies political parties target for legislative specialization. Bill initiatives do not represent the whole universe of legislative effort, and there are other legislative outputs that could serve as proxies of legislative effort. For instance, non-binding resolutions could signal concern for a particular policy jurisdiction. Nevertheless, bill initiatives are usually more costly in terms of both the time and staff resources needed to draft them. Furthermore, it is more likely that bill initiation becomes more salient and exposed by the media than non-binding resolutions. Moreover, at least a portion of bill initiatives expose the authors to the media (Callaghan and Schnell, 2010; Brunner, 2012). Considering these facts, I focus on the production of bill initiatives throughout this study as legislative parties or legislators employ them to publicize their preferences regardless of their approval or rejection. In fact, legislators or legislative parties could simply propose bills to signal policy preferences without much hope of having these bills passed.<sup>3</sup>

#### 2.1 Defining benefits

As stated above, parties could gain electoral support, at least in the short term, by targeting highly salient policies. Specifically, parties could seek to increase support from focusing on highly salient policy jurisdictions in congress that voters consider to be the most important areas for enhancing their welfare. In fact, previous studies find that *issue saliency* can have a relevant effect on voting behaviour (Rabinowitz *et al.*, 1982; Budge and Farlie, 1983; Edwards *et al.*, 1995; Fournier *et al.*, 2003; Bélanger and Meguid, 2008; Walgrave *et al.*, 2012; Lazarus, 2013; Klüver and Sagarzazu, 2016).

Another possible strategy that could alter the expected future electoral benefits party leaders estimate is accomplishing a strong reputation in a salient policy jurisdiction. As mentioned above, legislative parties might attempt to construct a reputation or claim issue ownership in order to maximize electoral support in the long term. The pursuit of reputation as a competent party to deal with problems in a particular policy jurisdiction might lead a party to spend legislative resources on highlighting its interest in a policy area (Green-Pedersen, 2010; Schröder and Stecker, 2018). This strategy could be conducive to increasing the degree of concentration of legislative specialization.

However, these short-term or long-term incentives might make parties within a party system adopt similar degrees of concentration of legislative specialization focusing on highly salient policy jurisdictions. Since all parties face the same potential benefits of legislative specialization, differences across parties in their levels of concentration of legislative specialization should depend on additional factors. Differences in informational advantages among parties – on which policy jurisdictions can provide greater electoral benefits – might help explain to some extent differences across parties in their concentration of legislative specialization. Nevertheless, as this study shows in the empirical section, the differences in the concentration of legislative specialization across parties are striking and can hardly

 $<sup>^{3}</sup>$ For more details on this issue, which is not the goal of this study, consult the work of Alemán *et al.* (2009). I acknowledge that the production of bill initiatives is not the only way to signal preferences or construct reputation.

be explained by only informational advantages, which might disappear fast as parties adjust their strategies. This leads to consideration of other types of incentives or constraints that might affect parties' structure of incentives of greater concentration of legislative specialization.

# 2.2 Defining marginal benefits and marginal costs

I assume that the marginal electoral benefits of greater concentration of legislative specialization for parties are positive but decrease as parties increase their concentration of legislative specialization. The heterogeneity in the voters' levels of satisfaction across policy jurisdictions helps explain why certain issues are preferred to be targeted by parties or legislators while others are discarded. As parties target the relatively more salient issues for short-term benefits (or maybe pursue ownership for longterm benefits), parties enhance electoral support. However, if parties persist in targeting a particular policy jurisdiction by proposing additional initiatives, the additional electoral rewards will still be positive, but lower. The assumption of diminishing marginal utility of individuals is commonly employed in neoclassical economics. Also called 'law of diminishing marginal utility', this law assumes that the first unit of consumption of a good or service yields more utility than the second and subsequent units (Nicholson, 2002). Previous empirical studies show that this axiom describes individuals' structure of preferences well (Horowitz et al., 2004). In this case, voters will reward additional parties' efforts in a particular policy jurisdiction, but with a decreasing trend. Once parties satisfy voters' most important demands regarding a particular issue, meeting other mandates will become less rewarding for parties. As the marginal benefits of targeting another policy area exceeds those of the targeted policy jurisdiction, it is expected that the party switches its attention to this new most attractive policy jurisdiction. Party leaders are likely to do so if they expect greater marginal benefits from targeting another policy jurisdiction. In case a particular policy was not subject to the assumption of diminishing marginal utility, all parties would propose all bill initiatives within the realm of only that policy jurisdiction. Certainly, this scenario is never observable. Overall, the theoretical assumption of diminishing marginal utility seems to be not only compelling, but also empirically valid after discarding the existence of these alternative scenarios.

Furthermore, greater concentration of legislative specialization and its potential benefits are not exempt of costs. Opportunity costs are especially relevant as we consider the individual agenda of legislators. The opportunity costs include the votes that could have been gained under the best alternative strategy. For instance, a bill initiative could either respond to the immediate demands of the constituencies of a legislator, or contribute to highlighting interest in a highly salient issue. Although these two objectives - legislator's or party's agenda - might not be totally incompatible, it is almost certain that opportunity costs will arise when a bill initiative cannot contribute to these two objectives equally. Another potential opportunity cost includes all votes that could have been gained if the party would have targeted the second most attractive policy area to increase electoral support. These opportunity costs comprise the electoral support that parties lose as they 'invest' more time, technical, and administrative resources to increase their concentration of legislative efforts in salient policy jurisdictions. Specifically, as legislators spend more time and effort to propose new bills that follow party leaders' directives, they might risk losing seats in the legislature because of their diminished responsiveness to the demands of their local constituencies. Initiatives designed by leaders are more likely to favour the construction of reputation or short-term support for the party than to respond to specific demands of citizens residing in specific electoral districts. However, a potential trade-off between strategies strengthening the support for the party label and party responsiveness to the demands of specific electoral circumscriptions must be taken into account by party leaders in order to maximize electoral rewards for the party as a whole. This potential trade-off between legislators' goals and partisan electoral benefits poses an optimizing problem to party leaders.

As legislators stop responding to more relevant demands from their local constituencies (due to an additional initiative aiming to pursue ownership or short-term support for the party as a whole), the opportunity costs of this additional initiative grow. That's why I assume in my theory of legislative

specialization the existence of positive and increasing marginal electoral costs. Total opportunity costs increase as the concentration of legislative specialization expands. If the marginal opportunity costs of greater concentration of legislative specialization were not positive and increasing, legislators would propose all bill initiatives following party leaders' directives. This type of scenario is not observable. Thus, this assumption is theoretically plausible after discarding the occurrence of the alternative scenario.<sup>4</sup>

To sum up, greater concentration of legislative specialization (at the expense of reducing response to localized demands) will increase the marginal and total electoral opportunity costs for the party while the electoral benefits of these strategies still grow – but less than the contributions of previous legislative efforts enhancing the concentration of legislative specialization. The presence of both increasing marginal electoral costs and decreasing marginal electoral benefits might produce a maximizing state in which the marginal benefits of the degree of concentration of legislative specialization equal its marginal costs. Party leaders could push their members to increase the concentration of their legislative specialization until achieving this state. In sum, this condition determines parties' optimal degree of concentration of legislative specialization.

### 2.3 The role of party size

I hypothesize that the largest and the smallest parties select the highest concentration of legislative specialization. First of all, changes in the number of legislators modify the cost structure of the concentration of legislative specialization. For instance, a decrease in the number of parties automatically increases the number of legislators of the average party. If party leaders did not push their members for producing more bill initiatives, each legislator would have to sacrifice less time and effort in complying with party leaders' directives since party leaders now can delegate tasks to a greater number of legislators.

This condition is certainly advantageous for parties. For instance, as parties become larger, the drafting of a certain number of additional bill initiatives – designed to boost the concentration of legislative specialization – might not require that each legislator sacrificed bill initiatives targeting local constituencies to achieve reelection (assuming that each legislator requires certain number of bill initiatives to get reelected). Hence, such increases in the number of legislators should lead to reductions in the opportunity costs of enhancing the concentration of legislative specialization. The decline in marginal and average costs,<sup>5</sup> as party size enhances, produces a new maximizing state in which greater concentration of legislative specialization is required to maximize electoral support. Party leaders are then incentivized to push their members in the legislative body to further the concentration of legislative specialization until this is achieved.<sup>6</sup>

The existence of declining average costs of the degree of concentration of legislative specialization as party size increases suggests the presence of informational scale economies. Under this scenario, the costs of constructing ownership of an issue or targeting a highly salient issue for the party might minimize when the number of legislators increases. If the number of legislators varies across parties, I should observe differences across parties in the optimal degree of concentration of legislative specialization since the structure of benefits and costs depends on the number of legislators. In sum, party size might matter for explaining differences in the concentration of legislative specialization across parties of different sizes, and consequently, across party systems with different degrees of fragmentation.

<sup>&</sup>lt;sup>4</sup>I acknowledge that I could also assume constant and positive marginal costs for the theory to successfully predict the same outcomes regarding the U-shaped relationship between party size and the concentration of legislative specialization.

<sup>&</sup>lt;sup>5</sup>This change could be represented by a shift downwards of the curve of marginal costs where the *X*-axis is the number of bill initiatives proposed to increase the concentration of legislative specialization, and the *Y*-axis represents the marginal gains and losses of electoral support (see online Figure 1 in Appendix A for a graphical representation). The online Appendices can be consulted in the author's webpage https://sites.google.com/site/aldofponceugolini/data.

<sup>&</sup>lt;sup>6</sup>See online Figure 1 in Appendix A for a graphical representation of this transition to a new optimizing state in which marginal benefits equal the modified levels of marginal costs.

#### 2.4 Becoming a niche party

The smallest parties – as measured by the number of their legislators – are likely to pursue ownership of a single issue; in addition, their small number of legislators might not be able to propose a sufficient number of bill initiatives to own more than a single policy jurisdiction. In this case, the concentration of legislative specialization might be relatively high for the smallest parties (e.g. niche parties). In fact, there is a considerable amount of literature on how beneficial it is for the smallest parties to become a 'niche party' by specializing on a single issue rather than trying to respond to citizens' demands in multiple policy jurisdictions (Meguid, 2005; Adams *et al.*, 2006, 2012; Ezrow, 2008; De Vries and Hobolt, 2012; Neundorf and Adams, 2018). These studies find that niche parties' leaders and their supporters pay more attention to issue debates and become more policy oriented than those of other parties (Neundorf and Adams, 2018). Overall, niche party's legislators might opt for concentrating most of their efforts on only one issue that could help define their party brand. Since the success of these parties may depend almost entirely on a single issue or a radical position on a topic, the relative value of creating a reputation in a single policy jurisdiction might increase considerably. As the marginal benefit of the issue that define the label of the niche party becomes relevant, the smallest parties are incentivized to increase their concentration of legislative specialization to secure these benefits and survival.<sup>7</sup>

Hence, I contend that a U-shaped curve defines the relationship between party size and the degree of concentration of legislative specialization. As party size decreases, the degree of concentration of legislative specialization diminishes due to the increase in the marginal opportunity costs. However, as party size becomes relatively very small, the benefits of greater concentration of specialization grow substantially due to the increase in the marginal benefits. Party leaders then seek to increase the party's degree of concentration. In sum, I hypothesize that the degree of concentration becomes the highest for the smallest and the largest parties. When party sizes are extreme – relatively small or large – parties might decide to optimize by choosing relatively high levels of legislative concentration. Medium-sized parties might select lower levels of concentration of legislative specialization. The following section justifies the case chosen to test the hypothesis of this study.<sup>8</sup>

## 3. The Latin American context

Party leaders play a decisive and important role in Latin American legislatures. John Carey describes this role as follows (2009: 25):

National party congresses invariably occur less frequently than legislative party group meetings, but national party executive committees generally have the authority to establish the party line... Many parties also retain disciplinary bodies, composed of national party leaders that are authorized to impose sanctions on legislators who break discipline on votes where a party line has been established.

 $<sup>^{7}</sup>$ This would produce a shift upwards of the curve of marginal benefits costs where the *X*-axis is the number of bill initiatives proposed to increase the concentration of legislative specialization, and the *Y*-axis represents the marginal gains and losses of electoral support. The result would be greater concentration of legislative specialization until marginal benefits equal marginal costs. See online Figure 2 in Appendix A for a graphical representation.

<sup>&</sup>lt;sup>8</sup>Online Appendix B (Figure 1 for niche parties, Figure 2 for medium size parties, and Figure 3 for large parties) summarize what would happen to the concentration of legislative specialization as party size grows. As explained above, parties target the policy jurisdictions that yield the greatest net marginal benefits (from left to right). When the ownership of a policy jurisdiction is achieved, the curve of net benefits shifts upwards due to this increase. This gain explains the rise of niche parties as they achieve the ownership of one policy jurisdiction. Once party size increases, the incentives related to the benefits of targeting another policy incentivizes parties to switch their strategy. Due to the law of diminishing marginal utility, this switch is likely to occur after some threshold. This would explain the decline in the concentration of legislative concentration as party size grows. However, as the resources for the party continue growing, it becomes feasible and electorally convenient to own more policy areas, parties might return to target a similar set of policy jurisdictions that provide the party with the greatest marginal benefits. If the ownership of these additional policy areas is (marginally) profitable, the party will continue targeting them.

Saiegh (2005: 23) notes that party leaders 'usually set policy agendas, nominate candidates<sup>9</sup> and monitor the work of elected representatives' in party meetings. These activities frequently take place in regularly scheduled meetings, which usually occur once a week. It is in these meetings that party policy is established, strategies for passing legislation or publicizing important matters are developed, and party and legislative leaders are selected.<sup>10</sup>

Latin American party leaders can strategically employ their prerogatives to reward obedient legislators or punish defiance. These instruments help party leaders influence legislators' behaviour, and consequently, help explain the relatively high levels of cohesion and party unity in the legislative arena.<sup>11</sup> For instance, Langston (2017) reports that party leaders in the legislature continued dominating decision-making in Mexico after the *Partido Revolucionario Institucional*'s defeat in the 2000 presidential election. Similar accounts have been reported for Brazil – a country in which electoral rules, federalism, and the inability of party leaders to nominate candidates tend to favour individual legislators' agendas over party agendas. Limongi and Cheibub Figueiredo (2007) and Figueiredo and Limongi (2000) argue that internal rules organize legislative work entirely around political parties, giving rise to highly centralized decision-making in the Brazilian congress. Although there is no clear consensus on the relative power of party leaders *vis-a-vis* that of legislators, there is general agreement that party leaders can exert a significant influence over legislators' actions in Latin America.

In sum, previous studies attribute an important and decisive role to party leaders in coordinating actions and strategies within their legislative groups in Latin American legislatures. Leaders contribute to solve effectively collective action problems within their parties (in the legislature). Based on these previous studies and the evidence from interviews with several legislators in the region,<sup>12</sup> it becomes reasonable to argue that party leaders are strong enough to influence legislators to propose some fraction of bill initiatives that could help parties develop reputations in specific policy jurisdictions.

In addition to the coordinating role that party leaders play in Latin American legislatures, I examine the effects of the determinants of the concentration of legislative specialization within the Latin American context because of several reasons. First, differences in party system fragmentation<sup>13</sup> produce significant variations in party size that make comparisons meaningful. Second, democratization and democratic consolidation are still recent phenomena in most countries of the region. During this period parties might be using selective legislative specialization to construct programmatic linkages with voters. Since the existence of strong programmatic bonds could encourage greater concentration of legislative specialization, analysing legislative specialization at the early stages of democratic consolidation helps minimize the potentially endogenous relationship between the concentration of legislative specialization and the existence of programmatic bonds between parties and citizens.

Third, Latin American countries share both common cultural roots and presidential systems,<sup>14</sup> which helps reduce the number of potential determinants explaining variation in parties'

<sup>&</sup>lt;sup>9</sup>Overall, primaries have been the exception rather than the rule in Latin America (Carey and Polga-Hecimovich, 2006; Aragón, 2009).

 $<sup>^{10}</sup>$ I confirmed the frequency and the goals of these meetings through extensive interviews with five legislators in Chile, two in Paraguay, and three in Uruguay. See online Appendix C for a summary of these interviews. The interviews provide evidence about the relevant coordinating roles leaders play within their legislative parties in Latin America.

<sup>&</sup>lt;sup>11</sup>Despite some variations across Latin American legislatures, overall levels of party unity are rather high (Saiegh, 2005; Alemán and Saiegh, 2007; Carey, 2009; Alemán, Ponce and Sagarzazu, 2011), which suggests effective party leadership.

<sup>&</sup>lt;sup>12</sup>None of these countries in which I interviewed legislators (Chile, Paraguay, and Uruguay) presents one of the highest scores of the centralization index. Most of the interviewed legislators admit the constant interference of party leaders to request the drafting of specific initiatives (see online Appendix C). It is then reasonable to assume that such interventions also occur in those countries with the most centralized parties.

<sup>&</sup>lt;sup>13</sup>See online Appendix D to consult these values and those of other relevant institutional traits included in the analyses as control variables. Clear differences in these characteristics are not easy to find in the post-Communist new democracies where all of them have been crippled by high party system fragmentation and high electoral volatility (Toka and Henjak, 2007).

<sup>&</sup>lt;sup>14</sup>According to the literature, presidentialism reduces party unity in congress (Carey, 2009). If the hypothesis of this study is verified, it is then reasonable to assume that the likelihood of observing this behaviour could be even greater in parliamentary systems or in the US congress.

concentration of legislative specialization. Finally, data on bill initiatives are available for several countries, which makes feasible the testing of the hypothesis of this study. The following section tests the validity of the hypothesis within the Latin American context.

#### 4. Data and empirical analyses

In order to test the effect of party size on the concentration of legislative specialization, I select 10 Latin American legislatures. The legislatures of these 10 countries – Brazil (2011–2014), Chile (1990–2014), Colombia (2010–2015), Dominican Republic (2002–2010), Guatemala (1996–2012), Mexico (2007–2012), Nicaragua (2006–2011), Paraguay (2003–2013), Peru (2001–2011), and Uruguay (1985–2010) – were strategically selected for several reasons. First, these countries register diverse scores of the concentration of legislative specialization and legislative fragmentation (see online Appendices D and E). Second, bill initiation data are available for these countries.<sup>15</sup> Third, the relative strength of party leaders varies across these countries, according to Jones's *Index of Centralization of Power in the Political Parties* (2005). For instance, while the Nicaraguan parties register one of the highest strength scores (13.5), the Chilean parties present one of the lowest scores in the region (it only reaches 9).

I employ bill initiation data for testing my hypothesis on the relationship between party size and the degree of concentration of legislative specialization.<sup>16</sup> Bill initiation data directly reveal which policy issues parties' legislators aim to emphasize. Another option might be the analysis of the composition of committee membership. Previous analyses of bill sponsorship show that committee membership constitutes the most important predictor of issue attention (Woon, 2011). However, parties might seek to channel relatively more initiatives through salient committees in which their membership presence is relatively less important. Moreover, the mere presence of legislators in a committee does not imply automatically effort made by the party. Considering these potential distortions in the measurement of issue attention employing committee membership data, I choose employing bill initiation data that directly capture the efforts that were made.

The number of bill initiatives proposed by legislators and scrutinized in the overall analysis totals 66,027 for all legislatures. I acknowledge that this dataset does not include the universe of bill initiatives in the region. However, this representative sample allows me to incorporate in the empirical analysis substantial variation regarding key institutional traits for this study such as legislative fragmentation, legislative capabilities, electoral volatility, district magnitude, and the degree of parties' centralization of power. It is then unlikely that the incorporation of data from other legislatures, which present similar characteristics to those selected in this study, change significantly the findings reported herein.

To assess the degree of concentration of legislative specialization across parties and across party systems, I classify these bill initiatives according to 12 possible policy jurisdictions.<sup>17</sup> This classification offers two advantages. First, there are committees representing these policy jurisdictions in all the congresses of my sample. Thus, it becomes possible to make comparisons across countries. Second, this classification includes all the most salient policy areas that the literature on issue ownership identifies to be relevant for voters.

To measure the concentration of legislative specialization of each party, I employ two indicators: the Gini coefficient and the Herfindahl index. These indices have different properties with different influences on the tails of the distribution. The Gini coefficient is a widely used measure of inequalities

<sup>&</sup>lt;sup>15</sup>Online Appendix E displays the legislatures employed in the empirical analyses and the sources of the data (bill initiatives coded), which are the official web pages of the lower houses of the selected countries.

<sup>&</sup>lt;sup>16</sup>Online Appendix E also shows: (1) the number of bill initiatives for each legislative period that I used in the empirical analyses; (2) the number of seats of each legislative party; and (3) the effective number of legislative parties for each legislative period.

<sup>&</sup>lt;sup>17</sup>Online Appendix F presents the list of issues. I only exclude bill initiatives regarding constitutional reforms since these changes do not clearly signal a policy jurisdiction which could be owned. I choose these jurisdictions based on the committees of the selected countries.

across units (Creedy, 1998). It ranges from 0 in cases of perfect equality across all units to 1 in cases of complete inequality. As used in this study, the Gini coefficient assesses the extent to which a party specializes across policy jurisdictions. Thus, a Gini coefficient of 0 indicates that a party allocated the same share of bill initiatives in every policy jurisdiction. A Gini coefficient of 1 means that a party assigned 100 per cent of its bill initiatives in one policy jurisdiction and 0 per cent in all the rest.

The Herfindahl index is an indicator of concentration, which is usually employed to measure the concentration of income and market share concentration (Hirschman, 1945; Herfindahl, 1950; Rhoades, 1993). It is calculated by squaring the shares (percentage) of bill initiatives allocated in every policy jurisdiction, and then summing the squares (Hirschman, 1945; Herfindahl, 1950; Rhoades, 1993). Since the index gives heavier weight to policy jurisdictions with larger shares than to policy jurisdictions with small shares, higher values of the Herfindahl index indicate greater concentration of legislative specialization (Hirschman, 1945; Herfindahl, 1950; Rhoades, 1993).<sup>18</sup>

To calculate these two measures, I first count how many bills legislators of a certain party presented. I exclude from this count all bill initiatives that legislators initiated to favour exclusively their constituencies in the electoral districts where they were elected. Legislators might employ these initiatives to respond to their constituencies' demands or to the push to build a favourable reputation of being competent to deal with particular issues in their districts.<sup>19</sup>

After counting the number of bill initiatives proposed in each policy jurisdiction by each party during a given year, I calculated the percentage of bills proposed by each party in each policy jurisdiction. Then I calculated the Gini coefficient and the Herfindahl index for every party and for every year.<sup>20</sup>

Considering that bounded dependent variables are not amenable to linear regression models, I fit a fractional response multilevel regression model with the automatic limits between 0 and 1 given the configuration of the Gini coefficient and the Herfindahl index. The fractional response regression model does not need specific data transformations at the corners and estimates the conditional expectation of the dependent variable given the predictors. The estimation of the parameters is based on a quasi-maximum-likelihood method that produces robust and relatively efficient estimates (Papke and Wooldridge, 1996). I take advantage of a multilevel regression employing a random intercept model for assessing the impact of the covariates on the dependent variables. In particular, I employ a multilevel model varying intercept to account for possible statistical dependence among observations belonging to the same legislature. Failure to cluster this type of data may result in underestimated standard errors and consequently to mistakes in the estimation of our inference analysis (Barcikowski, 1981; Steenbergen and Jones, 2002). To account for this, the use of multilevel modelling takes the hierarchical structure of the data into account by assuming random effects at each level of analysis. This provides a more conservative inference.

As the key independent variables to test my hypothesis regarding the effect of party size in congress, I employ two variables: the percentage of legislative seats and the square of the percentage of legislative seats each political party holds in congress. As the theory predicts, the results should indicate the presence of U-shaped relationship between party size and the concentration of legislative specialization: a

<sup>20</sup>I employed the *Stata* command 'ineqdec0' for calculating the Gini coefficient (Jenkins, 1999). I present the results of these calculations in the fourth column of online Appendix E. As online Appendix E shows, the differences in the degree of concentration of legislative specialization are significant across parties and countries.

<sup>&</sup>lt;sup>18</sup>The calculated Herfindahl values were divided by 10,000 to scale this dependent variable between the values of zero and one.

<sup>&</sup>lt;sup>19</sup>Consult Sulkin (2005) for further details on construction of legislators' reputations based on issues. Online Appendix E shows in the second column the number of bill initiatives proposed by every party and for every legislature, after eliminating the bill initiatives conceived to exclusively foster individual electoral gains. For cases in which the legislator might favour both the electoral circumscription in which she/he was elected and the party label, it is not possible to determine what the final goal of the legislator is. I acknowledge the existence of this limitation and the possible presence of measurement error. However, excluding those bills that benefit exclusively constituencies in a particular electoral district certainly ameliorates the measurement error problem (compared to including all bills). Unfortunately, there is not an available alternative way of eliminating completely this problem. I also acknowledge that I did not exclude those bills that seem to be of little relevance. Doing this could also introduce measurement error.

negative coefficient for the percentage of legislative seats and a positive coefficient for the square of the percentage of legislative seats.

As control variables, I first employ a measure of electoral volatility. To measure electoral volatility, I employ the electoral volatility indicator taken from Jones (2005), which is calculated based on the Pedersen index (Pedersen, 1979). Higher levels of this measure indicate higher levels of volatility. Greater electoral volatility might contribute to reducing the degree of concentration of legislative specialization. Expectations about a party's survival in the legislature might alter the expected electoral benefits of a reputation to deal with problems in a particular policy jurisdiction. For instance, in an electorally volatile context in which new parties emerge and old parties disappear, the expected electoral benefits of party reputation decrease. Therefore, parties might value these potential benefits relatively less. Since legislative parties expect to obtain electoral rewards from such reputations in the long term, they might be sensitive to changes in electoral volatility regarding their decisions on how many issues to highlight. As the expected electoral benefits of such reputation. For instance, if party leaders observe increases in electoral volatility, they might proceed to recalculate the future electoral benefits of such a reputation and re-prioritize the regional or local agenda of their legislators. Under these conditions, the concentration of legislative specialization might shrink.

I also employ a measure of legislative capabilities. Legislators' experience and the quality of staff could affect the ability of legislators and party leaders to alter the concentration of legislative specialization. As noted above, legislative parties might benefit from greater legislative capabilities because they could produce bill initiatives to signal their policy preferences to their constituencies. The congress capabilities index (of legislative capabilities), which is constructed by Sebastián Saiegh and reported by Stein *et al.* (2006), presents an important advantage. It focuses only on Latin American legislatures, which probably improves its accuracy as it analyses fewer legislatures with more similar characteristics. The index compares legislatures according to eight indicators, five quantitative and three qualitative. The first two quantitative indicators measure the confidence of citizens and business people in the performance of congress. The third indicator reflects the average years of legislator experience, and the fourth, the percentage of legislators with university education. Finally, the fifth quantitative indicators consider the strength of committees, whether the legislature is a good place to build a career, and technical expertise.

I also control for the relative power of party leaders within their organizations. If party leaders hold greater power within their organizations, they could reach more easily their optimal level of concentration of legislative specialization. Issue voting practices might lead empowered leaders to coordinate more successfully parties' strategies to increase the concentration of legislative specialization on fewer policy issues. Hence, this level of leaders' relative power might alter the selection of concentration of legislative specialization. For this purpose, I used Jones's *Centralization of Power Index* to account for these differences (2005). This index is based on five factors: leaders' right to select legislative candidates, design of the electoral system used for the election of legislative candidates, timing of presidential and legislative elections (whether or not these elections are concurrent), presence of autonomous regional leaders, and presidential primaries held.<sup>22</sup>

I also add in a control variable that indicates whether or not the party is the ruling one or is part of the ruling coalition. Given that Latin American executives are repositories of bureaucratic expertise and possess informational advantages with respect to legislators (Alemán and Tsebelis, 2016; Ponce, 2016), ruling parties might have access to these resources. These advantages over other

<sup>&</sup>lt;sup>21</sup>This would produce a shift downwards of the curve of marginal benefits costs where the *X*-axis is the number of bill initiatives proposed to increase the concentration of legislative specialization, and the *Y*-axis represents the marginal gains and losses of electoral support.

 $<sup>^{22}</sup>$ I provide further details on the construction and use of the volatility indices, the legislative capabilities indices, and the *centralization of power index* in online Appendix G.

legislative parties might affect the concentration of legislative specialization between those of the ruling parties and those of other parties. In fact, these advantages might make ruling parties optimize their concentration of legislative specialization under more favourable conditions. Finally, I include a measure of party system fragmentation to evaluate whether the effects of party size over the degree of concentration of legislative specialization hold, even after considering this indicator of competition. I rely on the Laakso-Taagepera index to construct this variable (Laakso and Taagepera, 1979).<sup>23</sup> The configuration of competition might help explain cross-country differences in how party leaders form their expectations about future gains and losses according to the competition they face.

For testing the theory regarding the existence of a U-shaped relationship between party size and the degree of concentration of legislative specialization, I employed four specifications (two fractional response models for the Gini coefficient and two additional one for the Herfindahl index). While the first model specification for each pair of model specifications does not include the party system fragmentation variable, the second one includes this additional covariate.

### 4.1. Results

I display results of the four model specifications in Table 1. The results confirm the existence of a U-shaped relationship between party size and the degree of concentration of specialization. All employed models provide evidence of this influence regardless of the employed measure of the degree of concentration of legislative specialization. Both the smallest and the largest parties tend to choose higher levels of concentration of legislative specialization. These types of parties embark more persistently in targeting fewer policy jurisdictions. Then, it is possible that these parties achieve ownership of policies more often, which might help them increase their electoral support. On the contrary, medium-sized parties seem to be more often inclined to concentrate less their legislative effort across policy jurisdictions; however, the policies targeted are likely to be salient and electorally profitable. The adoption of these different strategies might also produce divergent effects for internal politics. Party leaders might seek to accumulate greater power – and even achieve it – in the largest and smallest parties in order to facilitate the coordination of these optimizing strategies that involve greater concentration of legislative specialization. Further research shall provide information on the plausibility of these possible implications.

Regarding the control variables, some of them seem to matter for explaining the concentration of legislative specialization. First, legislative capabilities seem to matter for explaining the concentration of legislative specialization, but it is not statistically significant in all models. Enhancements in legislative capabilities are correlated with increases in the concentration of legislative specialization.

Furthermore, the relative power of leaders (centralization index) also seems to be positively associated with increases in the concentration of legislative specialization, but such association does not appear to be robust. Finally, reductions in party system fragmentation boost the concentration of legislative specialization. This result is robust even if the party size variables are included in the model. Further research shall focus on how variations in competition dynamics can also alter the concentration of legislative specialization.

In order to obtain an indication of the substantive size of the effects of size of the party, I report the expected values of the Gini coefficient and the Herfindahl index in Figure 1, as the values of party size vary. The other variables take on their mean values. As shown, these effects are not only statistically significant, but also meaningful. The values of the Gini coefficient and the Herfindahl index diminish substantially as the size of the party approaches an intermediate value (around 36 per cent of seats). Figure 1 confirms that the smallest and largest parties concentrate relatively more their legislative specialization.<sup>24</sup>

<sup>&</sup>lt;sup>23</sup>I show descriptive statistics of these variables and the sources of the data employed in online Appendices H and I.

<sup>&</sup>lt;sup>24</sup>I calculate all these values from models 2 and 4 of Table 1.

Type of model Models	Fractional response (multilevel) models			
	1	2	3	4
Variables	DV: Gini coefficient	DV: Gini coefficient	DV: Herfindahl index	DV: Herfindahl index
Constant	-1.05(1.23)	0.61 (1.06)	-2.09 (1.41)	-0.78 (1.47)
Testing size of the party				
Size of the party	-0.02*** (0.01)	-0.03*** (0.004)	-0.04*** (0.01)	-0.04*** (0.01)
Size of the party <sup>2</sup>	0.0003***	0.0004***	0.001***	0.001***
Control variables	(010001)	(010001)	(0.0002)	(010001)
Electoral volatility	-0.002 (0.01)	-0.01 (0.01)	-0.01 (0.01)	-0.01 (0.01)
Legislative capabilities	0.25	0.41***	0.33	0.44***
Centralization index	0.14***	0.08*	0.12**	0.08
Ruling party	-0.07	-0.05	-0.07 (0.08)	-0.05
Effective number of political parties	(0.00)	-0.12***	(0.00)	-0.10***
Number of observations	884	884	884	884
Log pseudo-likelihood	-570.19	-565.01	-544.53	-541.31

Table 1. Explaining the concentration of legislative specialization at the party level

\*\*\*Statistically significant at 1%; \*\*statistically significant at 5%; \*statistically significant at 10%.

## 4.2. Robustness checks

These results are robust to any change in the specification of the estimation. I adopt several strategies to evaluate the robustness of the results reported in Table 1. First, I take advantage of a fractional response regression model that employs party-clustered standard errors to examine if the inference is altered. As in all the robustness checks, each pair of models evaluates the two dependent variables of this study: Gini and Herfindahl indexes. The results from similar models to those of Table 1 confirm the validity of the hypothesis of this study.<sup>25</sup> Second, I use (party, legislative period, and country) fixed effects to examine whether or not party, legislative period, or country characteristics distort the significant effects of the key variables measuring the size of the party and the square of the size of the party on legislative specialization. I find that the effect of these variables is robust to this model specification.<sup>26</sup>

Third, I include a measure of district magnitude to evaluate whether or not this systemic trait of the electoral system alters the significant effects of party size on legislative specialization. District magnitude has been found to be a powerful predictor of party system fragmentation, being district magnitude and party system fragmentation positively related (Amorim Neto and Cox, 1997; Roberts Clark and Golder, 2006). Moreover, previous studies suggested that electoral systems, which are strongly determined by district magnitude, help explain why parties target a specific or broad range of voters (Cox, 1990; Myerson, 1993). This tendency might also create distortions in the degree of concentration of legislative specialization. The results of the econometric analysis indicate that increases in district magnitude are associated with reductions in the concentration of legislative specialization resembling those of party system fragmentation. However, this effect does not appear to be statistically significant,

<sup>&</sup>lt;sup>25</sup>The results of these models are displayed in online Appendix J.

<sup>&</sup>lt;sup>26</sup>The results of these models are displayed in Tables 1 and 2 in online Appendix K. While Table 1 displays specifications employing fractional response models, Table 2 presents those using fractional response multilevel models.



Figure 1. Expected values of the Gini coefficient and Herfindahl index as size of the party varies.

suggesting that district magnitude could be related to the concentration of legislative specialization through other effects not captured by legislative fragmentation.<sup>27</sup>

Fourth, I calculate a second Gini coefficient for every party by weighting every policy jurisdiction differently, employing the percentage of bill initiatives in every policy jurisdiction. This considers differences in the relevance of each policy jurisdiction in the legislative agenda to evaluate whether these variations could alter the validity of the hypothesis. The idea is to control for the biases that these variations might create for the estimations. For instance, some policy jurisdictions might be more

<sup>&</sup>lt;sup>27</sup>These results are displayed in Table 3 in online Appendix K. Furthermore, I employ seat volatility (also taken from Jones, 2005) instead of electoral volatility as a control variable. The results of these models are displayed in Table 4 in Online Appendix K.

important in some countries than in others (i.e. mining in Chile and Peru or security in Colombia). I rerun model 1 reported in Table 1 with this second type of Gini coefficient as the dependent variable. The results are very similar to those displayed in Table 1: both variables party size and party size square are statistically significant and present the expected sign.<sup>28</sup>

Fifth, I employ all bill initiatives including those affecting only the interests of a single electoral circumscription.<sup>29</sup> The new results remain very similar to those displayed in Table 1. Sixth, I run two specifications in which I exclude the smallest parties (parties holding control of 1 per cent or less of seats). Legislators in these parties might not be able to choose among different policy jurisdictions as their capabilities are substantially reduced. When the smallest parties are excluded from the sample, the results still confirm the hypothesis of this study.<sup>30</sup> Seventh, I run panel data models with random effects to determine whether or not time series effects could distort the results. I lag the dependent variable and include it as an additional covariate to capture dynamic effects in the specifications.<sup>31</sup> The key results of this study are robust to this alternative specification. Finally, I run a post-stratification econometric model designed to adjust the sampling weights considering the unequal sampling across countries. This method decreases the bias due to underrepresented countries in the sample. The results of this econometric model once again confirm the results previously reported.<sup>32</sup>

#### 5. Conclusions

This study demonstrates that the smallest and largest parties (relatively) tend to choose higher degrees of concentration of legislative specialization in order to maximize support. In a similar manner, medium-sized parties also optimize by selecting lower levels of concentration of legislative specialization. The selection of these strategies can produce important effects on the types of linkages parties hold with voters. Previous studies already report that in relatively more fragmented party systems, parties own fewer issues (usually at most one or two issues) than in two-party systems, and most of the policy issues are not owned by any party (Petrocik, 1996; Walgrave and De Swert, 2007; Bélanger and Meguid, 2008). My study sheds light on the causal mechanisms that could explain why this occurs. In particular, legislative fragmentation might help explain differences in the content of information that legislative bodies provide to voters. As the concentration of legislative specialization increases, voters might increasingly learn about programmatic differences among political parties. Under this scenario, credibility and generation of information are more likely to increase in a polity with programmatic parties; and this goal might become less costly under less fragmented party systems.

In particular, greater levels of concentration of legislative specialization could lead to a faster formation of programmatic linkages with voters. I recommend that further research be conducted to analyse whether this is the case. If further research confirmed this finding, and the formation of programmatic linkages was desirable, policymakers and politicians might evaluate the adoption of key institutional reforms to increase the degree of concentration of legislative specialization for achieving greater programmaticness. While niche parties – more commonly found in proportional systems – can survive, even grow, and respond to the specific demands of certain voters; the largest parties – more prevalent in majoritarian systems – could respond with greater emphasis to certain policy agendas. If the latter type of response was normatively preferable, reductions in district magnitude shall lead to

<sup>&</sup>lt;sup>28</sup>To see these results, consult online Appendix L. I display the results of these calculations in the third column of online Appendix E.

<sup>&</sup>lt;sup>29</sup>As explained above, these initiatives were excluded from the regression analysis whose results are shown in Table 1. I display these new findings in online Appendix M.

<sup>&</sup>lt;sup>30</sup>The results of these models are displayed in online Appendix N. In addition, I fit a multilevel Tobit model employing a maximum-likelihood estimation (with limits between 0 and 1) for those readers interested in seeing these results (although the data are not censored). The results of these models are displayed in online Appendix O.

<sup>&</sup>lt;sup>31</sup>Online Appendix P displays these results. Online Appendix Q includes country, party, and period effects. The key results remain unaltered.

<sup>&</sup>lt;sup>32</sup>Online Appendix R displays these results.

make this outcome relatively more common. An important benefit of low levels of district magnitude is that the presence of the largest parties might boost responsiveness in the long term, in case the formation of programmatic linkages was confirmed. However, some degree of responsiveness to shortterm demands might also be desirable, and those might be relatively lower for the largest parties as medium-sized parties might respond with greater emphasis to short-term and salient demands. These normative trade-offs deserve further analysis.

Supplementary material. The supplementary material for this article can be found at https://doi.org/10.1017/S1468109923000245 and https://dataverse.harvard.edu/dataset.xhtml?persistentId=doi:10.7910/DVN/BRZ5U7.

Acknowledgments. I thank Dr. Eduardo Alemán, Dr. Ernesto Calvo, Dr. Peter Kingstone, Dr. Susan Scarrow, and Dr. Jonathan Slapin as well as anonymous reviewers of the JJPS for their valuable comments on prior drafts of this article.

Competing interest. None.

### References

- Adams J, Clark M, Ezrow L and Glasgow G (2006) Are niche parties fundamentally different from mainstream parties? The causes and the electoral consequences of Western European parties' policy shifts, 1976–1998. American Journal of Political Science 50, 513–529. doi.org/10.1111/j.1540-5907.2006.00199.x. Available at https://scholar.google.com/scholar?cluster=15004618843589596405&hl=en&as\_sdt=0,5
- Adams J, Ezrow L and Debra L (2012) Partisan sorting and niche parties in Europe. West European Politics 35, 1272–1294. doi.org/10.1080/01402382.2012.713746. Available at https://scholar.google.com/scholar?cluster=193197787350046618&hl= en&as\_sdt=0,5
- Aldrich J and Griffin J (2003) The presidency and the campaign: creating voter priorities in the 2000 election. In Nelson M (ed.), *The Presidency and the Political System*, 7th edn. Washington, DC: Congressional Quarterly, pp. 239–256. Available at https://scholar.google.com/scholar?hl=en&as\_sdt=0%2C5&q=Aldrich%2C+J+and+Griffin%2C+J+%282003%29+The +Presidency+and+the+Campaign%3A+Creating+Voter+Priorities+in+the+2000+Election&btnG=
- Alemán E and Saiegh SM (2007) Legislative preferences, political parties, and coalition unity in Chile. *Comparative Politics* 39, 253–272. Available at https://www.jstor.org/stable/20434040; https://scholar.google.com/scholar?cluster=137561396007370 22188&hl=en&as\_sdt=0,5
- Alemán E and Tsebelis G (2016) Legislative Institutions and Lawmaking in Latin America. Oxford: Oxford University Press. Available at https://scholar.google.com/scholar?cluster=16891254950889671835&hl=en&as\_sdt=0,5
- Alemán E, Calvo E, Jones MP and Kaplan N (2009) Comparing cosponsorship and roll call ideal points. Legislative Studies Quarterly 34, 87–116. 10.3162/036298009787500358. Available at https://scholar.google.com/scholar?cluster=11983370 525018949381&hl=en&as\_sdt=0,5
- Alemán E, Ponce A and Sagarzazu I (2011) Legislative parties in volatile, nonprogrammatic party systems: The Peruvian case in comparative perspective. *Latin American Politics and Society* 53, 57–81. 10.1111/j.1548-2456.2011.00125. Available at https://scholar.google.com/scholar?oi=bibs&hl=en&cluster=16234033082433256216
- Amorim Neto O and Cox G (1997) Electoral institutions, cleavage structures, and the number of parties. *American Journal of Political Science* **41**, 149–174. Available at https://www.jstor.org/stable/2111712; https://scholar.google.com/scholar? cluster=7965686784263901793&hl=en&as\_sdt=0,5
- Aragón F (2009) Candidate Nomination Procedures and Political Selection: Evidence from Latin American Parties. LSE STICERD Research Paper No. EOPP003. Available at https://ssrn.com/abstract=1546897; https://scholar.google.com/ scholar?cluster=6710123194717554696&hl=en&as\_sdt=0,5
- Barcikowski R (1981) Statistical power with group mean as the unit of analysis. *Journal of Educational Statistics* 6, 267–285. Available at https://www.jstor.org/stable/1164877; https://scholar.google.com/scholar?cluster=9548504648506137394&hl=en&as\_sdt=0,5
- Baron D and Ferejohn J (1989) Bargaining in legislatures. *The American Political Science Review* 83, 1181–1206. Available at https://www.jstor.org/stable/1961664; https://scholar.google.com/scholar?cluster=6352259558710593888&hl=en&as\_sdt=0,5
- Bélanger E and Meguid B (2008) Issue salience, issue ownership and issue-based vote choice: evidence from Canada. Electoral Studies 27, 477–491. 10.1016/j.electstud.2008.01.001. Available at https://scholar.google.com/scholar?cluster= 11056698614782864043&hl=en&as\_sdt=0,5
- Bos L, Lefevere J, Thijssen R and Sheets P (2016) The impact of mediated party issue strategies on electoral support. *Party Politics* 23, 760–771. 10.1177/1354068815626. Available at https://scholar.google.com/scholar?cluster=1793642698884235 9538&hl=en&as\_sdt=0,5
- Brunner M (2012) Parliaments and Legislative Activity: Motivations for Bill Introduction. Wiesbaden: Springer Science & Business Media. Available at https://scholar.google.com/scholar?cluster=1635854987141004349&hl=en&as\_sdt=0,5

- Budge I and Farlie D (1983) Explaining and Predicting Elections Issue Effects and Party Strategies in Twenty-Three Democracies. London: George Allen & Unwin.
- Callaghan K and Schnell F (2010) Assessing the democratic debate: how the news media frame elite policy discourse. *Political Communication* 18, 183–213. 10.1080/105846001750322970. Available at https://scholar.google.com/scholar? cluster=1617397024213869901&hl=en&as\_sdt=0,5
- Carey J (2009) Legislative Voting and Accountability. Cambridge: Cambridge University Press. Available at https://scholar.google.com/scholar?cluster=7971849217950120354&hl=en&as\_sdt=0,5
- Carey J and Polga-Hecimovich J (2006) Primary elections and candidate strength in Latin America. *Journal of Politics* 68, 530–543. 10.1111/j.1468-2508.2006.00443.x. Available at https://scholar.google.com/scholar?cluster=24683534039549645 92&hl=en&as\_sdt=0,5
- Clarke H, Sanders D, Stewart M and Whiteley P (2004) Political Choice in Britain. Oxford: Oxford University Press, pp. 79–129. Available at https://scholar.google.co.uk/scholar?hl=en&as\_sdt=0,5&cluster=4667144407987379757
- Cox G (1990) Centripetal and centrifugal incentives in electoral systems. American Journal of Political Science 34, 903–935. 10.2307/2111465. Available at https://scholar.google.co.uk/scholar?hl=en&as\_sdt=0%2C5&q=Centripetal+and+centrifugal +incentives+in+electoral+systems&btnG=
- Cox G and McCubbins M (1993) Legislative Leviathan: Party Government in the House. Berkeley: University of California Press. Available at https://scholar.google.co.uk/scholar?cluster=12650997926078938825&hl=en&as\_sdt=0,5
- Creedy J (1998) The Dynamics of Inequality and Poverty: Comparing Income Distributions. Northampton, MA: E. Elgar. Available at https://scholar.google.com/scholar?cluster=8137183521898997176&hl=en&as\_sdt=0,5
- Damore D (2004) The dynamics of issue ownership in presidential campaigns. *Political Research Quarterly* 57, 391–397. 10.2307/3219849. Available at https://scholar.google.com/scholar?cluster=17593191676761631305&hl=en&as\_sdt=0,5
- De Vries C and Hobolt S (2012) When dimensions collide: the electoral success of issue entrepreneurs. European Union Politics 13, 246–268. 10.1177/14651165114347. Available at https://scholar.google.com/scholar?cluster=10907206354513 134546&hl=en&cas\_sdt=0,5
- Edwards G, Mitchell W and Welch R (1995) Explaining presidential approval: the significance of issue salience. *American Journal of Political Science* **39**, 108–134. 10.2307/2111760. Available at https://scholar.google.com/scholar?cluster=18358633557819118215&hl=en&as\_sdt=0,5
- Ezrow L (2008) Research note: on the inverse relationship between votes and proximity for niche parties. European Journal of Political Research 47, 206–220. 10.1111/j.1475-6765.2007.00724.x. Available at https://scholar.google.com/scholar? cluster=6428197261007170110&hl=en&as\_sdt=0,5
- Figueiredo A and Limongi F (2000) Presidential power, legislative organization, and party behavior in Brazil. *Comparative Politics* **32**, 151–170. 10.2307/422395. Available at https://scholar.google.com/scholar?cluster=4325735354632030035&hl= en&as\_sdt=0,5
- Fournier P, Blais A, Nadeau E, Gidengil E and Nevitte N (2003) Issue importance and performance voting. *Political Behavior* 25, 51–67. Available at https://www.jstor.org/stable/3657314; https://scholar.google.com.mx/scholar?cluster= 2704293165379379004&hl=en&as\_sdt=0,5&as\_vis=1
- Gilligan T and Krehbiel K (1987) Collective decision-making and standing committees: an informational rationale for restrictive amendment procedures. *Journal of Law, Economics, and Organization* 3, 287–335. Available at https://www.jstor.org/stable/764831; https://scholar.google.com/scholar?cluster=1192428332826686806&hl=en&as\_sdt=0,5
- Green-Pedersen C (2010) Bringing parties into parliament: the development of parliamentary activities in Western Europe. Party Politics 16, 347–369. Available at 10.1177/1354068809341057 https://scholar.google.com/scholar?cluster= 10254115497756470814&hl=en&as\_sdt=0,5
- Green-Pedersen C and Mortensen P (2015) Avoidance and engagement: issue competition in multiparty systems. *Political Studies* 63, 747–764. 10.1111/1467-9248.12121. Available at https://scholar.google.com/scholar?cluster=74070232951860 60644&hl=en&as\_sdt=0,5
- Green J and Hobolt S (2008) Owning the issue agenda. Party strategies and vote choices in British elections. *Electoral Studies* 27, 460–476. 10.1016/j.electstud.2008.02.003. Available at https://scholar.google.com/scholar?cluster=144468508157490 46313&hl=en&as\_sdt=0,5
- Green J and Jennings W (2019) Party reputations and policy priorities: how issue ownership shapes executive and legislative agendas. *British Journal of Political Science* 49, 443–466. 10.1017/S0007123416000636. Available at https://scholar.google.com.mx/scholar?cluster=2189950143770085555&hl=en&as\_sdt=0,5&as\_vis=1
- Hamm K, Hedlund R and Post S (2011) Committee specialization in U.S. State legislatures during the 20<sup>th</sup> century: do legislatures tap the talents of their members? *State Politics and Policy Quarterly* 11, 299–324. Available at https://www.jstor.org/stable/41575827; https://scholar.google.com/scholar?cluster=12988682160625176824&hl=en&as\_sdt=0,5
- Hayes D (2005) Candidate qualities through a partisan lens: a theory of trait ownership. American Journal of Political Science
  49, 908–923. 10.1111/j.1540-5907.2005.00163.x. Available at https://scholar.google.com/scholar?cluster=17659901615980
  200631&hl=en&sa\_sdt=0,5
- Herfindahl O (1950) Concentration in the Steel Industry (Doctoral Dissertation), Columbia University. Available at https:// scholar.google.com/scholar?hl=en&as\_sdt=0%2C5&q=Concentration+in+the+steel+industry&btnG=

- Hirschman A (1945) National Power and the Structure of Foreign Trade. Berkeley: University of California Press. Available at https://scholar.google.com/scholar?cluster=13642290884391487542&hl=en&as\_sdt=0,5
- Holian D (2004) He's stealing my issues! Clinton's crime rhetoric and the dynamics of issue ownership. *Political Behavior* 26, 95–124. Available at https://www.jstor.org/stable/4151362; https://scholar.google.com/scholar?cluster=1334382955530054 22&hl=en&as\_sdt=0,5
- Horowitz J, List J and McConnell KE (2004) Diminishing marginal value. Paper presented at the American Agricultural Economics Association Annual Meeting, Denver, Colorado. 10.22004/ag.econ.20297. Available at https://scholar.google.com/scholar?cluster=17290255846398825502&hl=en&as\_sdt=0,5
- Jenkins S (1999) INEQDEC0: Stata module to calculate inequality indices with decomposition by subgroup. Statistical Software Components S366007, Boston College Department of Economic. Available at https://scholar.google.com/scholar?hl=en&as\_sdt=0%2C5&q=Stephen+P.+Jenkins%2C+1999.+%22INEQDEC0%3A+Stata+module+to+calculate +inequality+indices+with+decomposition+by+subgroup&btnG=
- Jones M (2005) The role of parties and party systems in the policymaking process. Paper prepared for Inter-American Development Bank Workshop on State Reform, Public Policies and Policymaking Processes, February 28–March 2, Washington, DC. Available at https://scholar.google.com/scholar?cluster=15031927104955788216&hl=en&as\_sdt=0,5
- Klüver H and Sagarzazu I (2016) Setting the agenda or responding to voters? Political parties, voters and issue attention. West European Politics 39, 380–398. 10.1080/01402382.2015.1101295. Available at https://scholar.google.com/scholar? cluster=3473682147305954327&hl=en&ca\_sdt=0,5
- Krehbiel K (1991) Information and Legislative Organization. Ann Arbor: The University of Michigan Press. Available at https://scholar.google.com/scholar?cluster=14120666270118410063&hl=en&as\_sdt=0,5
- Laakso M and Taagepera R (1979) The 'effective' number of parties: a measure with application to West Europe. *Comparative Political Studies* 12, 3–27. 10.1177/001041407901200101. Available at https://scholar.google.com/scholar? cluster=13870868923305472042&hl=en&as\_sdt=0,5
- Langston J (2017) Democratization and Authoritarian Party Survival: Mexico's PRI. Oxford: Oxford University Press. Available at https://scholar.google.com/scholar?cluster=15383924655104837250&hl=en&as\_sdt=0,5
- Lazarus J (2013) Issue salience and bill introduction in the house and senate. Congress and the Presidency 40, 215–229. 10.1080/07343469.2013.822944. Available at https://scholar.google.com/scholar?cluster=14041819900481096771&hl=en& as\_sdt=0,5
- Lefevere J, Tresch A and Walgrave S (2015) Introduction: issue ownership. West European Politics 38, 755-760. 10.1080/ 01402382.2015.1039375. Available at https://scholar.google.com/scholar?cluster=15423759735378935800&hl=en&as\_sdt=0,5
- Limongi F and Cheibub Figueiredo A (2007) The budget process and legislative behavior: individual amendments, support for the executive and government programs. *World Political Science Review* 3, 1–34. 10.2202/1935-6226.1024. Available at https://scholar.google.com/scholar?cluster=12977929225790172048&hl=en&as\_sdt=0,5
- Mainwaring S (ed.) (2018) Party Systems in Latin America: Institutionalization, Decay, and Collapse. New York: Cambridge University Press. Available at https://scholar.google.com/scholar?hl=en&as\_sdt=0%2C5&q=Party+Systems+in+Latin +America%3A+Institutionalization%2C+Decay%2C+and+Collapse+book&btnG=
- Meguid B (2005) Competition between unequals: the role of mainstream party strategy in niche party success. American Political Science Review 99, 347–359. Available at https://www.jstor.org/stable/30038944; https://scholar.google.com/scholar?cluster=1535228314770554387&hl=en&as\_sdt=0,5
- Myerson R (1993) Incentives to cultivate favored minorities under alternative electoral systems. American Political Science Review 87, 856–869. Available at https://www.jstor.org/stable/2938819; https://scholar.google.com/scholar?cluster=146 17074556088816740&hl=en&as\_sdt=0,5
- Nadeau R, Blais A, Gidengil E and Nevitte N (2001) Perceptions of party competence in the 1997 election. In Thorburn H and Whitehorn A (eds), *Party Politics in Canada*. Toronto: Prentice-Hall, pp. 413–430. Available at https://scholar.google.com/scholar?hl=en&as\_sdt=0%2C5&q=Perceptions+of+Party+Competence+in+the+1997+Election&btnG=
- Neundorf A and Adams J (2018) The micro-foundations of party competition and issue ownership: the reciprocal effects of citizens' issue salience and party attachments. *British Journal of Political Science* 48, 385–406. 10.1017/S0007123415000642. Available at https://scholar.google.com/scholar?cluster=11345853226995129953&hl=en&as\_sdt=0,5
- Nicholson W (2002) Microeconomic theory. Basic principles and extensions. South-Western; Thomson.
- Norris P, Curtice J, Sanders D, Scammell M and Semetko H (1999) On Message: Communicating the Campaign. London: Sage. Available at https://scholar.google.com/scholar?cluster=7048681286292616118&hl=en&as\_sdt=0,5
- Papke L and Wooldridge J (1996) Econometric methods for fractional response variables with an application to 401(k) plan participation rates. *Journal of Applied Econometrics* 11, 619–632 10.1002/(SICI)1099-1255(199611)11:6<619::AID-JAE418>3.0.CO;2-1. Available at https://scholar.google.com/scholar?cluster=5263271696875280565&hl=en&as\_sdt=0,5
- Pedersen M (1979) The dynamics of European party systems: changing patterns of electoral volatility. *European Journal of Political Research* 7, 1–26. 10.1111/j.1475-6765.1979.tb01267.x. Available at https://scholar.google.com/scholar?hl=en&as\_sdt= 0%2C5&q=The+Dynamics+of+European+Party+Systems%3A+Changing+Patterns+of+Electoral+Volatility&btnG=
- Petrocik J (1996) Issue ownership and presidential elections, with a 1980 case study. American Journal of Political Science 40, 825–850. Available at https://www.jstor.org/stable/2111797; https://scholar.google.com/scholar?cluster=46440110598009 7158&hl=en&as\_sdt=0,5

- Petrocik J, Benoit W and Hansen G (2003) Issue ownership and presidential campaigning, 1952–2000. Political Science Quarterly 118, 599–626. 10.1002/j.1538-165X.2003.tb00407.x. Available at https://scholar.google.com/scholar?cluster= 8930374849571906186&hl=en&as\_sdt=0,5
- Ponce AF (2016) Strong presidents, weak parties, and agenda control: Lawmaking in Democratic Peru. In Alemán E and Tsebelis G (eds), *Legislative Institutions and Lawmaking in Latin America*. Oxford: Oxford University Press. Available at https://scholar.google.com/citations?view\_op=view\_citation&hl=en&user=cZl28P4AAAAJ&citation\_for\_view=cZl28P4AAAAJ;j3f4tGmQtD8C
- Rabinowitz G, Prothro J and Jacoby W (1982) Salience as a factor in the impact of issues on candidate evaluation. *Journal of Politics* 44, 41–63. Available at https://www.jstor.org/stable/2130283; https://scholar.google.com/scholar?cluster=4443668 328308752452&hl=en&as\_sdt=0,5
- Rhoades S (1993) The Herfindahl–Hirschman index. Federal Reserve Bulletin, March, 188–189. Available at https://scholar.google.com/scholar?cluster=4816391282417053115&hl=en&as\_sdt=0,5
- Roberts Clark W and Golder M (2006) Rehabilitating Duverger's theory: testing the mechanical and strategic modifying effects of electoral laws. *Comparative Political Studies* **39**, 679–708. 10.1177/0010414005278420. Available at https://scholar.google.com/scholar?cluster=17993876316700223387&hl=en&as\_sdt=0,5
- Rohde D (1991) Parties and Leaders in the Post-Reform House. Chicago: The University of Chicago Press.
- Saiegh S (2005) The role of legislatures in the policymaking process. Prepared for delivery at the Workshop on State Reform, Public Policies and Policymaking Processes, Inter-American Development Bank, Washington, D.C. Available at https:// scholar.google.com/scholar?hl=en&as\_sdt=0%2C5&q=The+Role+of+Legislatures+in+the+Policymaking+Process&btnG=
- Schröder V and Stecker C (2018) The temporal dimension of issue dimension. *Party Politics* 24, 708–718. 10.1177/ 1354068817693474. Available at https://scholar.google.com/scholar?cluster=11950266664500475720&hl=en&as\_sdt=0,5
- Steenbergen M and Jones B (2002) Modeling multilevel data structures. American Journal of Political Science 46, 218–237. Available at https://www.jstor.org/stable/3088424 https://scholar.google.com/scholar?cluster=12004403903485358539&hl= en&as\_sdt=0,5
- Stein E, Tommasi M, Echebarría K, Lora E and Payne M (eds) (2006) The Politics of Policies. Economic and Social Progress in Latin America. Cambridge: David Rockefeller Center. Available at https://scholar.google.com/scholar?cluster=1850 413924871526471&hl=en&as\_sdt=0,5
- Sulkin T (2005) Issue Politics in Congress. Cambridge: Cambridge University Press. Available at https://scholar.google.com/ scholar?cluster=9885065771502247723&hl=en&as\_sdt=0,5
- Toka G and Henjak A (2007) Party systems and voting behavior in the Visegrad countries 15 years after the transition. In Saradín P and Bradová E (eds), *Visegrad Votes: Parliamentary Elections 2005–2006*. Olomuc: Palacky University Press, pp. 210–244. Available at https://scholar.google.com/scholar?cluster=1322764533486915522&hl=en&as\_sdt=0,5
- van der Brug W (2004) Issue ownership and party choice. *Electoral Studies* 23, 209–233. 10.1016/S0261-3794(02)00061-6. Available at https://scholar.google.com/scholar?cluster=10221539755915584054&hl=en&as\_sdt=0,5
- Walgrave S and De Swert K (2004) The making of the (issues of the) Vlaams Blok: the media and the success of the Belgian extreme-right party. *Political Communication* 21, 479–500. 10.1080/10584600490522743. Available at https://scholar. google.com/scholar?cluster=15790521261419341158&hl=en&as\_sdt=0,5
- Walgrave S and De Swert K (2007) Where does issue ownership come from? From party or from the media? Issue-identification in Belgium, 1991–2005. The Harvard International Journal of Press and Politics 12, 37–66. 10.1177/ 1081180X06297572. Available at https://scholar.google.com/scholar?cluster=13458168252788722410&hl=en&as\_sdt=0,5
- Walgrave S, Lefevere J and Nuytemans M (2009) Issue ownership stability and change: how political parties claim and maintain issues through media appearances. *Political Communication* 26, 153–172. 10.1080/10584600902850718. Available at https://scholar.google.com/scholar?cluster=15029462189713838170&hl=en&as\_sdt=0,5
- Walgrave S, Lefevere J and Tresch A (2012) The associative dimension of issue ownership. *Public Opinion Quarterly* 76, 771–782. 10.1093/poq/nfs023. Available at https://scholar.google.com/scholar?cluster=12269062865775773126&hl=en& as\_sdt=0,5
- Weingast BR and Marshall WL (1988) The industrial organization of congress; or, Why legislatures, like firms, are not organized as markets. *Journal of Political Economy* 96, 132–163. Available at https://www.jstor.org/stable/1830714; https:// scholar.google.com/scholar?cluster=11511531026978890666&hl=en&as\_sdt=0,5
- Woon, J (2011) Issue attention and legislative proposals in the U.S. Senate. *Legislative Studies Quarterly* **34**, 29–54. Available at https://www.jstor.org/stable/20680226; https://scholar.google.com/scholar?cluster=13640054613183731481&hl=en& as\_sdt=0,5

**Cite this article:** Ponce AF (2023). Explaining the allocation of legislative specialization. *Japanese Journal of Political Science* **24**, 391–408. https://doi.org/10.1017/S1468109923000245